
CENWP-OP-B                                                                                                                                  11 March, 2002

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 10 of 2002, which covers the period 03 through 09 March 2002.
1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 117.8 to 128.1kcfs.  Daily project tailwater elevation averaged 11.4’ to 12.1’ msl.  Daily project forebay elevation averaged 73.7’ to 75.6’ msl.  River temperature averaged 45 to 46° F.  Daily turbidity averaged 5.5 to  >7’ on the secchi disk.

1.2 Spill Bays 18 and 1 are closed for spill deflector work.  

1.3 Unit 5 remains out of service for turbine rehab.  Unit 17 remains out of service for VBS slot modification work.   Unit 10 remains out of service for its five-year overhaul.  Unit 11 returned to service at 2300 on 04 March.  
Units out of service for dive operations:

04 March- Units 12 and 13 (0636-1330), Unit 14 (1018- 1330), Unit 15 (1118- 1330), Unit 16 (1330- 1618), and Unit 18 (1330- 1700)
Units out of service for trash rack raking: 

05 March- Unit 11 (0700- 0806 and from 0836- 1406), Unit 14 (0818- 0830)
Units out of service for research transducer installation:

06 March- Unit 12 (0718- 1306), Unit 13 (1306- 1548)

07 March- Unit 14 (0654- 0836), Unit 15 (0836- 1030), Unit 16 (1030- 1242), Unit 18 (1242-1606)
Units out of service for STS or ESBS installation:

08 March- Unit 18 (0700-0936), Unit 16 (0936-1100), Unit 14 (1100-1306), Unit 13 (1306-1418), and Unit 12 (1418-1548)


09 March- Unit 11 (0700-0930), Unit 15 (1112-1436)

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby from approximately 0000-0300 nightly, if drawdown exceeds 1.0’.  Fish Unit 1 remains out of service for winter maintenance.  As coordinated with NMFS, F2 is supplying all attraction flow at this time. (See Misc. 5.3).
1.5 Gatewell Drawdown was measured on 06 March.  All units were within criteria.
2.  ACTIVITIES. 
2.1 AWS Closures.  FV5-3, FV5-4, and  FV5-9 returned to service at 1554 on 08 March.

2.2 STS/VBS Inspections.  All STSs have been installed for fish passage season.

2.3 Dewatering and Fish Salvage.  No dewatering or fish salvage activities for this reporting week.

3.  RESEARCH.

3.1 WDFW Fish Counting.  24-hour video count ongoing at the Bradford Island and Washington Shore ladders.

3.2 PSMFC Adult PIT.  Final walkthrough was at 1000 on 5 March.  Non in-water work ongoing. 

3.3 PSMFC/NMFS Juvenile PIT.  Monitoring to start with Spring Creek release on 11 March. 

3.4 WDFW Winter Steelhead tagging.  Trapping and tagging of steelhead continues.

3.5 NMFS Lamprey.  Tracking continues. 

3.6 PSMFC JMF.  Smolt monitoring is scheduled to begin on 11 March.

3.7 NMFS Saltwater rearing.  Set up underway, start up is expected for April.

4.  FISHWAYS. 
4.1 Project biologists inspected 04-07 March this reporting week.

4.2 Fish Passage Plan Observations.

DATE
LOCATion
FPP Deviation
CAUSE
RESPONSE
TR/WAR

3/4-7
PH1 PCC velocity
Velocity <1.5fps on the electronic meter.
Unknown- the rest of the ladder is in criteria.
No action at this time.


3/3-3/9
PH2 PCC velocity meter.
Electronic meter not yet installed.
Divers weren’t able to install parts during 19 February dive.
The transducer will be installed during 13 March dive.


3/3-6
PH2 DSM
DSM out of service
Construction activities
The walk through for in water work wasn’t until 6 March.


3/3,8,9

Biologists did not inspect 7 days per week.
Personnel shortage.
Job advertised.


3/7
Weir 37
Reading > or < 1.0’
A dampener needs to be installed in the stillwell.
Electricians will fix during 13 March dive.


4.3 Fish Ladder Temperatures.  Temperatures are measured 3 weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.   Temperatures are in Celsius.

Date
Wash. Shore
Bradford Is.

3/3



¾
5.2
5.2

3/5
5.3
5.4

3/6
5.3
5.4

3/7
5.3
5.4

3/8
5.2
5.4

3/9
5.1
5.3

4.4 Sensor Calibration.  Calibration was checked on 04 March this reporting week.

4.5 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week. 

5. MISCELLANEOUS

5.1 To aid spawning Chum below Bonneville, a minimum tailwater restriction of 11.4’ went into effect at 1054 on 28 November.  The restriction will continue until further notice.

5.2 As coordinated with NMFS, the ITC will not be used for AWS water.  F2 is being operated to maintain a 1.5’ differential.  In an effort to reduce the debris load during daylight hours, the fish unit is put on standby from 1800 until 0500 to float trash.

5.3 At 1030 on 03 March, the Second Powerhouse became priority.

5.4 DSM1 had to be turned south from 0955 on 05 March until 1553 on 06 March due to a failure of the small trash sweep.

5.5 DSM2 returned to service at 1725 on 07 March.  The hydrocannons were in operation by 0724 on 08 March.

 DEBORAH L. CHENOWETH

                                 

                                                             Operations Project Manager
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